Author's response to reviews

Title: Comparison of Published Orthopaedic Trauma Trials Following Registration in Clinicaltrials.gov

Authors:

Rajiv Gandhi (rajiv.gandhi@uhn.on.ca)
Meryam Jan (m.jan91@hotmail.com)
Holly N Smith (hollysmith183@hotmail.com)
Nizar N Mahomed (nizar.mahomed@uhn.on.ca)
Mohit Bhandari (bhandam@mcmaster.ca)

Version: 2 Date: 19 July 2011

Author's response to reviews: see over
To the BMC Musculoskeletal Reviewers,

A number of requested revisions were made to the original manuscript.

The manuscript was changed to a ‘type 3 correspondence.’ The abbreviation ‘CTG’ was removed in the abstract and changed to ‘ClinicalTrials.gov.’ The ‘Level of Evidence’ section in the abstract was modified to ‘Keywords.’ A List of Abbreviations Used section was added to the text. The ‘Background’ section of the manuscript was changed to an ‘Introduction’ as part of the text.

Thank you.

Sincerely,
Meryam Jan